Author's response to reviews

Title: What do bereaved parents want from professionals after the sudden death of their child? A structured narrative review of the literature.

Authors:

Joanna J Garstang (joanna.j.garstang@warwick.ac.uk)
Frances E Griffiths (F.E.griffiths@warwick.ac.uk)
Peter D Sidebotham (P.sidebotham@warwick.ac.uk)

Version: 4  Date: 29 September 2014

Author's response to reviews: see over
The Editor  
BMC Pediatrics  
29 September 2014  

Dear Sir,

Manuscript number 4916675851295954 - What do bereaved parents want from professionals after the sudden death of their child? A systematic review of the literature

I have revised the manuscript as requested. The revisions are as follows:

1. Page 2, line 29. The abstract refers to the paper as a systematic review
2. Tables 2, 3 and 4. These have been amended to show the year of publication.

Yours faithfully

Dr Joanna Garstang  
MBChB MSc MRCPCH  
NIHR Doctoral Research Fellow  
Division of Mental Health and  
Wellbeing  
Tel: +44 (0)24 7651 3144  
Email: j.garstang@warwick.ac.uk

Medical School Building  
The University of Warwick  
Coventry CV4 7AL, United Kingdom  
Tel: +44 (0)24 7657 4880  
Fax: +44 (0)24 7652 8375